Author’s response to reviews

Title: Broadening the phenotype of the TWNK gene associated Perrault syndrome

Authors:

Bálint Fekete (fba8713@gmail.com)
Klára Pentelényi (pentelenyi.klara@gmail.com)
Gábor Rudas (rudas.gabor@semmelweis-univ.hu)
Anikó Gál (galaniko77@gmail.com)
Zoltán Grosz (groszz@hotmail.com)
Anett Illés (netty.illes@gmail.com)
Idris Jimoh (idris.jimoh@yahoo.com)
Gábor Csukly (csugab@yahoo.com)
Andor Domonkos (molneur@gmail.com)
Mária Molnár (molnarmj@gmail.com)

Version: 2 Date: 13 Nov 2019

Author’s response to reviews:

Dear Arwa Tawfiq

Thank you for your letter. We corrected the manuscript accordingly:

1. Now Bálint Fekete is the corresponding author
2. We modified the format to the proposed one
3. We modified the "Consent for publication" part as requested
4. We modified the "Ethics approval and consent to participate" part as requested
5. We added an "Abbreviations" section as requested
6. We modified the "Availability of data and materials" section and added a website link to request the genomic data

7. We modified the "Competing interests" section as requested

8. We modified the "Role of funding body" section as requested

9. We updated the "Authors' contributions" section as requested and added some additional information to clarify the roles

10. We added the citation to the text for "Figure 1B"

11. We added the citation to the text for "Table 1"

12. We removed the requested attachments

13. We uploaded the manuscript as requested

Thank you for your observation and guidance

Bálint Fekete